Author’s response to reviews

Title: Antenatal Care and Mothers’ Education Improved Iron-Folic Acid Adherence at Denbiya District Health Centers, Northwest Ethiopia: Using pills count method.

Authors:

Missa Tarekegn (missatarekegn@gmail.com)
Mamo Wubshet (mamo_wubshet@yahoo.com)
Azeb Atenafu (bemnyaz@gmail.com)
Terefe Derso (dersotere@gmail.com)
Abere Azagew (wabere@ymail.com)

Version: 2 Date: 10 May 2019

Author’s response to reviews:

Dear editors/reviewers

Greetings

You are recommended to restructured the abstract to Background, Methods, Results, Conclusion and Trial registration. The trial registration is not applicable for this study that is why the authors omit it. If It is a clinical trial or systematic review the authors are obliged to put it at the abstract session.